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1) Switcher Submission (Peter Fischer): 
Switcher B has been submitted February 15, 2010 and should be back in 2 ½-3  months. Pefi stressed the fact that the peak current is 100mA. Therefore additional decoupling capacitors (1 nF) are needed on the balcony. The balcony layout foresees them, but specific types and their impact on the material have to be determined. It might be possible that part of the decoupling can be implemented in later versions of the chip.
Rainer asked whether these large peak currents could cause problems with electromigration. To be checked. 

It was asked whether this high currents lead to increased power consumption: No, the average power will be as specified.

The chip will be made on 600µm thick substrate (exact value to be confirmed). It could be thinned if this is deemed necessary.

Peter will prepare a list of voltage supplies and ratings for the services meeting next Monday.

2) QED background generators (Christian Kiesling): 

Christian had a meeting with the authors of KoralW in Krakow. Surprisingly the code seems to work very well at the low q² we are interested in, but the events weights need readjustment. This being done the rates and spectra compare well with other programs. No the expected occupancy is in the order of 0.07-0.1%. Special background runs with random triggers are planned at Belle to measure this background (or at least give an upper limit). 
Burkard asked whether this runs can be made with reduced B-field in order to increase the sensitivity to lower pt. To be checked.

Background files will be provided for simulation studies

3) Thermo-mechanical mockup (Martin Ritter):

Martin showed nice pictures of the mockup made by Charlie Ackermann. It is now preferred to fixe the modules with screws instead of springs and one of the supports slide in order to compensate for thermal expansions of the beam pipe. There are pros and cons for this, one of the few disadvantages is that the structure then has to be made in steel, not copper.
Laci made the proposal to separate the two make the two halves completely independent, not gluing the modules to ladders (possible since the inner layer modules will be split as well). Module assembly will be much easier then. Disadvantages: larger gap between modules, vibrations, difficult mounting? 

The proposal will be studied.

4) B ->D*+D*- 0ptimisation Studies (Oksana Brovchenko):
Oksana showed first results with simulated B->D*D* events using the Mocca/Marlin code. This channel can be used to study the effect of parameter variations on the reconstruction of low pt tracks. Among other things she showed the effect of increasing the sensor thickness from 50µm to 75µm on the impact parameter resolution for low pt, which is surprisingly small. Probably this reflects the material of the beam pipe. An study of the effect of the beam pipe material should be made.

Some suggestions: efficiency plots should be made including geometrical acceptance. A new release of the code should be used which has more realistic parameters. It was suggested to wait for the next release Kolja is preparing.

5) KEK News (Shuji Tanaka):
Shuji reported on the recent PAC review at KEK. Concerning the PXD the reviewers asked for a simulation of the effect of the material of the support structure on the background of the outer detectors. Also the estimated occupancy should be clarified.
The PXD-TDR will be reviewed in May (23/24?).
A new KEKB schedule was shown, the PXD installation date has shifted to the 2nd half of FY2013.

Radiation tests with FPGAs and opt. links died at 64krad. This may be a problem. We need to know the radiation level at the location of our opt. links and need to find out the radiation tolerance of the Gigabyte links we need to use.

In the B2GM there will be a common IP/PXD/SVD session (1st April 10:00) for mechanics and services and the standard PXD session at 14:00. It was proposed to add a common session on backgrounds at 9:00 (or whenever possible)

In the discussion it was agreed to study the PXD performance as function of the occupancy. This could give some indications till generators of beam related background become available.

6) DHP Submission (Hans Krüger):
The DHP test chip has been submitted in February. Fabrication needs 12 weeks plus time for C4 bump placement (should be fast because it is made in house by IBM). It turned out that wire bond pads and C4 ponds are incompatible, so the chip will not be wire-bondable. The bump bond adaptor needed will be produced in Munich

Hans asked if the test chip for clustering has been submitted: Andreas Wassatsch: Not yet.

7) TDR (Christian Kiesling)

The PXD TDR draft has been delivered in time and is posted on the Twiki. Few comments (6) were received so far. A new version with the DHP section will be released today. Everybody is encouraged to read the document and comment. Christian will prepare a new version with line numbers. PDF files with comments are equally welcome. The editors will then implement the comments and release the final version in two weeks. Zdenek will organize that the English is checked by a native speaker. 
8) AOB:
The action list was not discussed for lack of time. Everybody is invited to check and feedback the status of the actions he is concerned. 
Next Meeting: March 23, 2010: Preparation of the B2GM
Action list (not discussed,  additions –red- by HGM):
Proposed Actions (March 2, 2010)

1. (A1.1 from A1 and A13) EOS Geometry:  Since the first layer will be split as well, the constrains on the length of the EOS are now more relaxed and can be revised. Define module envelop for clearances (height of active and passive components and connectors, wire bonds etc). 

Charly Ackermann can do the drawings, but needs dimensions of components
2
.  (A3) Freeze module lengths (as soon as IR geometry is fixed) (TC)
3.   (A4) Backside connection of DEPFETs: Laci, Rainer

Favourite option: punch through. Tests started by Christian Koffmane, looks promising. Backup solution in work
4. (A8.1 from A7 and A8) Revise functionality of DHH assuming a patch panel can be used to have it farer from the IP. Finalize dimensions and location of these boxes.

5. (A9) Göttingen to check the cable layout by Hans-Günther (On Wiki). 

6. (A10) Conceptual design of air cooling: MPI, Valencia, Karlsruhe

7. (A11.1 from A10, A11 and A16) Design common cool (and dry) volume for PXD and SVD air cooling system. Define pipe dimensions (Carlos Marinas and I. Fall from Vienna)
8. (A14) Carlos Lacasta will discuss C3F8 cooling with ATLAS experts; Carlos is in contact with Georg Viehauser and will travel to Oxford. 

9. (A15.1 from A15, A14) Choose coolant; Water, CO2, C3F8 (MPI, Valencia, Karlsruhe, KEK). 

Follow up common effort with CERN on CO2 cooling. 

Carlos Lacasta in contact with Gerog Viehauser (Oxford) on C3F8

8.   (A41) Valencia will buy a CO2 compressor for tests.

10. (A17) Simulations should include background. Since background calculations are not yet available, study performance as function of occupancy (Christian). In work. Various backgrounds can be mixed in MC events. Results beginning of March.
11. (A18) Order SOI wafers (March, 2010): Tracit contacted, waiting for reply. Tracit will not produce SOI, propose Lethi. Need to look for and qualify alternative suppliers. (A19) Look for other sources of SOI wafers.
12. (A19) Finalize parameters relevant for SOI order (sensor thickness) before March 2010. Need to decide in December: Wait for A18
13. (A20) Mock up of IR and inner PXD/SVD: IR task force: First space problem has to be solved and drawings need to be prepared

14. (A21) PXD DAQ should be integrated in common DAQ in FY2010 (Giessen). In work. Need to decide whether ATCA or eventbuilder farm will be used.

15. (A22) Organize first meeting of ID task force (Ushiroda, Tsuboyama, Friedl, Moser, Shuji Tanaka, NN)

16. (A23) Measure sensitivity of system to noise and pickup. Stefan Rummel will work on pickup measurements

17. (A24) Grounding scheme (to avoid loops) (Santiago). First recommendations available (Santiago). Will be discussed in special PS & Services meeting early March.

18. (A25) Clean up TWIKI pages and interface to Belle II Twiki (?) drop
19. (A26) Support for PXD on beam pipe (Shuji). Shuji will prepare schematic drawings and send them to Charlie.

20. (A27) Need information on NIEL and SEU. NIEL damage might be severe. Background estimate needed and irradiations with electrons are needed.
Special run with separated beams planned.
21. (A28) Irradiate DEPFETs with 10MeV electrons at ELSA, Bonn (Ritter, Müller). Scheduled for April

22. (A29) Do we need hardware alignment monitor? Contact Ivan Villa

23. (A30) Hans-Günther to buy a Phenix Audio System with echo suppression - done
24. (A33) Feedback to Shuji on PXD assembly, installation and service routing
25. (A34) Produce dummy modules asap (Laci). Simple dummies are ready. Thinned dummies in produced, need to be cut. – done- 
26. (A35) Check legal constraints for high pressure pipes in Japan (Shuji): Shuji will find out and post it on our TWiki. 

27. (A36) Resolution studies should also look at the digitisation noise of the ADC 

28. (A37) Continue with pair background studiers. Look in ‘Traps’ program. There exists another program used by Masako – check!
29. (A38)New organisation of B2GM Meeting. PXD Session for PXD specific topics. Services and IR related issues in a common session with SVD and IR (Shuji) done. Another session on background should be organized (-> A48)
30. (A39) Organise meeting on power supplies, regulators, DHH & repeater early March (Hans-Günther). Done, Scheduled March 8, 2010.
31. (A40) Do some voltages need current sinks?

New

A42 Frank Simon to find out about rad hard Gigabit optical links (2.3.2010)

A43 Get information on the expected radiation level at the locations of the opt. links (2.3.2010)

A44 Peter Fischer to check effect of electromigration in switcher B (2.3.2010)

A45 Peter Fischer to write a list of supply voltages and ratings of switcher B (2.3.2010)

A46 Christian Kiesling to provide files with new KoralW background events to the simulation group (2.3.2010)

A47 Check whether the scheduled background runs at Belle can be done with a reduced B-field (2.3.2010)

A48 Study effect of beam pipe material on impact parameter resolution (2.3.2010)

A49 Shuji Tanaka to organize a parallel session on background during the B2GM in March/April (2.3.2010)

